Author's response to reviews

Title: The C allele of the JAK2 rs4495487 is an additional candidate locus that contributes to myeloproliferative neoplasm predisposition in the Japanese population

Authors:

Junko H Ohyashiki (junko@hh.iij4u.or.jp)
Masayuki Yoneta (dm106141@cc.seikei.ac.jp)
Hisashi Hisatomi (hisatomi@st.seikei.ac.jp)
Tamiko Iwabuchi (t-dorachan@jcom.home.ne.jp)
Tomohiro Umezu (t_umezu@tokyo-med.ac.jp)
Kazuma Ohyashiki (ohyashik@rr.iij4u.or.jp)

Version: 6 Date: 16 January 2012

Author's response to reviews:

The BMC Series Editorial Production Team,

Thank you for your e-mail on January 12, 2012.

I am sending herewith the formatted manuscript entitled “The C allele of JAK2 rs4495487 is an additional candidate locus that contributes to myeloproliferative neoplasm predisposition in the Japanese population.” We corrected a couple of typographical errors in the manuscript.

Yours sincerely,

Junko Ohyashiki